Author’s response to reviews

Title: Building COPD care on shaky ground - healthcare professional perspectives. A mixed methods study in Swedish primary care

Authors:
Sara Lundell (sara.lundell@umu.se)
Malin Tistad (malin.tistad@umu.se)
Börje Rehn (borje.rehn@umu.se)
Maria Wiklund (maria.e.wiklund@umu.se)
Åsa Holmner (asa.holmner@vll.se)
Karin Wadell (karin.wadell@umu.se)

Version: 2 Date: 02 Jun 2017

Author’s response to reviews:

Reviewer 2: Thank you for your valuable comment on our revision. After reading this section again we fully understood your comment and agreed that the section needed re-writing to be fully understood.

We have now tried to make it more clear, and we hope that you now think it make sense. Otherwise we of course will look over it once again.

Please see our revision in the discussion, page 24, lines 557-561.